Author’s response to reviews

Title: Daily hemodialysis practices in Australia/New Zealand and in France: a comparative cohort study

Authors:

Adelaide Pladys (adelaide.pladys@ehesp.fr)
Sahar Bayat (Sahar.Bayat-Makoei@ehesp.fr)
Cecile Couchoud (cecile.couchoud@biomedecine.fr)
Cecile Vigneau (cecile.vigneau@chu-rennes.fr)
Stephen McDonald (stephen@anzdata.org.au)

Version: 2 Date: 29 Mar 2019

Author’s response to reviews:

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

   > The manuscript has been entirely revised by a native English speaker.

2. Please clarify in the Ethics approval and consent to participate section why verbal consent was obtained. Did the ethics committee approve this procedure?

   > Patients included received an information leaflet before giving a verbal consent to participate. We modified the Ethics approval and consent to participate section as below:

   “All involved subjects received an information leaflet before giving their verbal consent to participate. This procedure was approved by the ethics committee. This study was approved by the French Biomedecine Agency and by the ANZDATA registry. This study included patients’ data that were anonymized and de-identified directly in the database and before the extraction for analysis.”
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

> The version uploaded is the clean one without any tracked changes or highlights. Each figure has been uploaded in a separate word document (Figures 1 to 3). All supplementary files have been uploaded in a unique word document (Additional file 1).